Author's response to reviews

Title: Pantoea spp. causing early onset neonatal sepsis (EONS): an unusual case report

Authors:

    Shreekant Tiwari (drshreekant@rediffmail.com)
    Sibashanker Beriha (drsiba@rediffmail.com)

Version: 4 Date: 2 January 2015

Author's response to reviews:

Respected sir
Herewith we are re submitting the manuscript to you for kind consideration of peer review. Care checklist of information and Timeline sample format is attached.

With regards

Dr Shreekant Tiwari